Author's response to reviews

Title: Criteria-based audit to improve a district referral system in Malawi: a pilot study

Authors:

Eugene J Kongnyuy (kongnyuy@gmail.com)
Grace Mlava (gracemlava90@gamil.com)
Nynke van den Broek (broek@liverpool.ac.uk)

Version: 4 Date: 15 September 2008

Author's response to reviews:

Dear Editors,
We have addressed the all editorial comments - we replaced Introduction by Background in the Abstract and Body of the manuscript and removed any figure and table placement text.

We hope the manuscript meet the requirement for publication.

Sincerely,

Eugene Kongnyuy